\begin{thebibliography}{}
\expandafter\ifx\csname natexlab\endcsname\relax\def\natexlab#1{#1}\fi
\providecommand{\url}[1]{\href{#1}{#1}}

\bibitem[{{Astropy Collaboration} {et~al.}(2018){Astropy Collaboration},
  {Price-Whelan}, {Sip{\H{o}}cz}, {G{\"u}nther}, {Lim}, {Crawford}, {Conseil},
  {Shupe}, {Craig}, {Dencheva}, {Ginsburg}, {Vand erPlas}, {Bradley},
  {P{\'e}rez-Su{\'a}rez}, {de Val-Borro}, {Aldcroft}, {Cruz}, {Robitaille},
  {Tollerud}, {Ardelean}, {Babej}, {Bach}, {Bachetti}, {Bakanov}, {Bamford},
  {Barentsen}, {Barmby}, {Baumbach}, {Berry}, {Biscani}, {Boquien}, {Bostroem},
  {Bouma}, {Brammer}, {Bray}, {Breytenbach}, {Buddelmeijer}, {Burke},
  {Calderone}, {Cano Rodr{\'\i}guez}, {Cara}, {Cardoso}, {Cheedella}, {Copin},
  {Corrales}, {Crichton}, {D'Avella}, {Deil}, {Depagne}, {Dietrich}, {Donath},
  {Droettboom}, {Earl}, {Erben}, {Fabbro}, {Ferreira}, {Finethy}, {Fox},
  {Garrison}, {Gibbons}, {Goldstein}, {Gommers}, {Greco}, {Greenfield},
  {Groener}, {Grollier}, {Hagen}, {Hirst}, {Homeier}, {Horton}, {Hosseinzadeh},
  {Hu}, {Hunkeler}, {Ivezi{\'c}}, {Jain}, {Jenness}, {Kanarek}, {Kendrew},
  {Kern}, {Kerzendorf}, {Khvalko}, {King}, {Kirkby}, {Kulkarni}, {Kumar},
  {Lee}, {Lenz}, {Littlefair}, {Ma}, {Macleod}, {Mastropietro}, {McCully},
  {Montagnac}, {Morris}, {Mueller}, {Mumford}, {Muna}, {Murphy}, {Nelson},
  {Nguyen}, {Ninan}, {N{\"o}the}, {Ogaz}, {Oh}, {Parejko}, {Parley}, {Pascual},
  {Patil}, {Patil}, {Plunkett}, {Prochaska}, {Rastogi}, {Reddy Janga},
  {Sabater}, {Sakurikar}, {Seifert}, {Sherbert}, {Sherwood-Taylor}, {Shih},
  {Sick}, {Silbiger}, {Singanamalla}, {Singer}, {Sladen}, {Sooley},
  {Sornarajah}, {Streicher}, {Teuben}, {Thomas}, {Tremblay}, {Turner},
  {Terr{\'o}n}, {van Kerkwijk}, {de la Vega}, {Watkins}, {Weaver}, {Whitmore},
  {Woillez}, {Zabalza}, \& {Astropy Contributors}}]{Ast18}
{Astropy Collaboration}, {Price-Whelan}, A.~M., {Sip{\H{o}}cz}, B.~M., {et~al.}
  2018, \aj, 156, 123

\bibitem[{{Astudillo-Defru} {et~al.}(2017){Astudillo-Defru}, {Delfosse},
  {Bonfils}, {Forveille}, {Lovis}, \& {Rameau}}]{Def17}
{Astudillo-Defru}, N., {Delfosse}, X., {Bonfils}, X., {et~al.} 2017, \aap, 600,
  A13

\bibitem[{Barnes {et~al.}(2005)Barnes, Cameron, Donati, James, Marsden, \&
  Petit}]{barnes2005dependence}
Barnes, J., Cameron, A.~C., Donati, J.-F., {et~al.} 2005, Monthly Notices of
  the Royal Astronomical Society: Letters, 357, L1

\bibitem[{Barnes {et~al.}(2004)Barnes, James, \&
  Cameron}]{barnes2004differential}
Barnes, J., James, D., \& Cameron, A.~C. 2004, Monthly Notices of the Royal
  Astronomical Society, 352, 589

\bibitem[{{Baroch} {et~al.}(2020){Baroch}, {Morales}, {Ribas}, {Herrero},
  {Rosich}, {Perger}, {Anglada-Escud{\'e}}, {Reiners}, {Caballero},
  {Quirrenbach}, {Amado}, {Jeffers}, {Cifuentes}, {Passegger}, {Schweitzer},
  {Lafarga}, {Bauer}, {B{\'e}jar}, {Colom{\'e}}, {Cort{\'e}s-Contreras},
  {Dreizler}, {Galad{\'\i}-Enr{\'\i}quez}, {Hatzes}, {Henning}, {Kaminski},
  {K{\"u}rster}, {Montes}, {Rodr{\'\i}guez-L{\'o}pez}, \&
  {Zechmeister}}]{Baroch2020}
{Baroch}, D., {Morales}, J.~C., {Ribas}, I., {et~al.} 2020, \aap, 641, A69

\bibitem[{{Bernstein} {et~al.}(2003){Bernstein}, {Shectman}, {Gunnels},
  {Mochnacki}, \& {Athey}}]{Ber03}
{Bernstein}, R., {Shectman}, S.~A., {Gunnels}, S.~M., {Mochnacki}, S., \&
  {Athey}, A.~E. 2003, in Society of Photo-Optical Instrumentation Engineers
  (SPIE) Conference Series, Vol. 4841, \procspie, ed. M.~{Iye} \& A.~F.~M.
  {Moorwood}, 1694--1704

\bibitem[{{Berta} {et~al.}(2012){Berta}, {Irwin}, {Charbonneau}, {Burke}, \&
  {Falco}}]{Ber12}
{Berta}, Z.~K., {Irwin}, J., {Charbonneau}, D., {Burke}, C.~J., \& {Falco},
  E.~E. 2012, \aj, 144, 145

\bibitem[{{Browning} {et~al.}(2010){Browning}, {Basri}, {Marcy}, {West}, \&
  {Zhang}}]{Browning2010}
{Browning}, M.~K., {Basri}, G., {Marcy}, G.~W., {West}, A.~A., \& {Zhang}, J.
  2010, \aj, 139, 504

\bibitem[{Cameron {et~al.}(2017)Cameron, Dikpati, \& Brandenburg}]{Cam17}
Cameron, R.~H., Dikpati, M., \& Brandenburg, A. 2017, Space Science Reviews,
  210, 367.
\newblock \url{https://doi.org/10.1007/s11214-015-0230-3}

\bibitem[{{Catalano} \& {Marilli}(1983)}]{Catalano1983}
{Catalano}, S., \& {Marilli}, E. 1983, \aap, 121, 190

\bibitem[{{Chabrier} \& {Baraffe}(1997)}]{Cha97}
{Chabrier}, G., \& {Baraffe}, I. 1997, \aap, 327, 1039

\bibitem[{{Charbonneau}(2014)}]{Cha14}
{Charbonneau}, P. 2014, \araa, 52, 251

\bibitem[{{Delfosse} {et~al.}(1998){Delfosse}, {Forveille}, {Perrier}, \&
  {Mayor}}]{Del98}
{Delfosse}, X., {Forveille}, T., {Perrier}, C., \& {Mayor}, M. 1998, \aap, 331,
  581

\bibitem[{{Dikpati} \& {Charbonneau}(1999)}]{Dik99}
{Dikpati}, M., \& {Charbonneau}, P. 1999, \apj, 518, 508

\bibitem[{Earl {et~al.}(2021)Earl, Tollerud, Jones, Kerzendorf, O'Steen, Busko,
  shaileshahuja, D'Avella, Robitaille, Ginsburg, Homeier, Sipőcz, Averbukh,
  Ogaz, Geda, Tocknell, Davies, Günther, Cherinka, Barbary, Foster,
  Droettboom, Torres, Bray, Casey, Teuben, Crawford, Ferguson, Lim, \&
  Cruz}]{Nic21}
Earl, N., Tollerud, E., Jones, C., {et~al.} 2021, astropy/specutils: v1.3,
  vv1.3,  Zenodo, doi:10.5281/zenodo.4987675.
\newblock \url{https://doi.org/10.5281/zenodo.4987675}

\bibitem[{{Gastine} {et~al.}(2014){Gastine}, {Yadav}, {Morin}, {Reiners}, \&
  {Wicht}}]{Yad13}
{Gastine}, T., {Yadav}, R.~K., {Morin}, J., {Reiners}, A., \& {Wicht}, J. 2014,
  \mnras, 438, L76

\bibitem[{Hall {et~al.}(2007)Hall, Lockwood, \& Skiff}]{Hal07}
Hall, J.~C., Lockwood, G.~W., \& Skiff, B.~A. 2007, The Astronomical Journal,
  133, 862.
\newblock \url{https://doi.org/10.1086%2F510356}

\bibitem[{{Hawley} {et~al.}(1996){Hawley}, {Gizis}, \& {Reid}}]{Hawley96}
{Hawley}, S.~L., {Gizis}, J.~E., \& {Reid}, I.~N. 1996, \aj, 112, 2799

\bibitem[{{Hawley} \& {Pettersen}(1991)}]{Haw91}
{Hawley}, S.~L., \& {Pettersen}, B.~R. 1991, \apj, 378, 725

\bibitem[{{Houdebine} {et~al.}(2017){Houdebine}, {Mullan}, {Bercu}, {Paletou},
  \& {Gebran}}]{Hou17}
{Houdebine}, E.~R., {Mullan}, D.~J., {Bercu}, B., {Paletou}, F., \& {Gebran},
  M. 2017, \apj, 837, 96

\bibitem[{{Irwin} {et~al.}(2011){Irwin}, {Berta}, {Burke}, {Charbonneau},
  {Nutzman}, {West}, \& {Falco}}]{Irw11}
{Irwin}, J., {Berta}, Z.~K., {Burke}, C.~J., {et~al.} 2011, \apj, 727, 56

\bibitem[{{Kelson}(2003)}]{Kel03}
{Kelson}, D.~D. 2003, \pasp, 115, 688

\bibitem[{{Kelson} {et~al.}(2000){Kelson}, {Illingworth}, {van Dokkum}, \&
  {Franx}}]{Kel00}
{Kelson}, D.~D., {Illingworth}, G.~D., {van Dokkum}, P.~G., \& {Franx}, M.
  2000, \apj, 531, 159

\bibitem[{{Kesseli} {et~al.}(2018){Kesseli}, {Muirhead}, {Mann}, \&
  {Mace}}]{Kesseli2018}
{Kesseli}, A.~Y., {Muirhead}, P.~S., {Mann}, A.~W., \& {Mace}, G. 2018, \aj,
  155, 225

\bibitem[{{Lehtinen} {et~al.}(2020){Lehtinen}, {K{\"a}pyl{\"a}}, {Olspert}, \&
  {Spada}}]{Leh20}
{Lehtinen}, J.~J., {K{\"a}pyl{\"a}}, M.~J., {Olspert}, N., \& {Spada}, F. 2020,
  arXiv e-prints, arXiv:2007.00040

\bibitem[{Lovis {et~al.}(2011)Lovis, S{\'e}gransan, Mayor, Udry, Benz, Bertaux,
  Bouchy, Correia, Laskar, Curto, {et~al.}}]{Lov11}
Lovis, C., S{\'e}gransan, D., Mayor, M., {et~al.} 2011, Astronomy \&
  Astrophysics, 528, A112

\bibitem[{{Magaudda} {et~al.}(2020){Magaudda}, {Stelzer}, {Covey}, {Raetz},
  {Matt}, \& {Scholz}}]{Magaudda2020}
{Magaudda}, E., {Stelzer}, B., {Covey}, K.~R., {et~al.} 2020, \aap, 638, A20

\bibitem[{{Mamajek} \& {Hillenbrand}(2008)}]{Mamajek2008}
{Mamajek}, E.~E., \& {Hillenbrand}, L.~A. 2008, \apj, 687, 1264

\bibitem[{{Mann} {et~al.}(2016){Mann}, {Gaidos}, {Mace}, {Johnson}, {Bowler},
  {LaCourse}, {Jacobs}, {Vanderburg}, {Kraus}, {Kaplan}, \& {Jaffe}}]{Man16}
{Mann}, A.~W., {Gaidos}, E., {Mace}, G.~N., {et~al.} 2016, \apj, 818, 46

\bibitem[{{Medina} {et~al.}(2020){Medina}, {Winters}, {Irwin}, \&
  {Charbonneau}}]{Med20}
{Medina}, A.~A., {Winters}, J.~G., {Irwin}, J.~M., \& {Charbonneau}, D. 2020,
  \apj, 905, 107

\bibitem[{{Middelkoop}(1982)}]{Middelkoop1982}
{Middelkoop}, F. 1982, \aap, 107, 31

\bibitem[{Newton {et~al.}(2017)Newton, Irwin, Charbonneau, Berlind, Calkins, \&
  Mink}]{Newton2017}
Newton, E.~R., Irwin, J., Charbonneau, D., {et~al.} 2017, The Astrophysical
  Journal, 834, 85

\bibitem[{{Newton} {et~al.}(2016){Newton}, {Irwin}, {Charbonneau},
  {Berta-Thompson}, {Dittmann}, \& {West}}]{Newton2016}
{Newton}, E.~R., {Irwin}, J., {Charbonneau}, D., {et~al.} 2016, \apj, 821, 93

\bibitem[{{Newton} {et~al.}(2018){Newton}, {Mondrik}, {Irwin}, {Winters}, \&
  {Charbonneau}}]{Newton2018}
{Newton}, E.~R., {Mondrik}, N., {Irwin}, J., {Winters}, J.~G., \&
  {Charbonneau}, D. 2018, \aj, 156, 217

\bibitem[{{Noyes} {et~al.}(1984){Noyes}, {Hartmann}, {Baliunas}, {Duncan}, \&
  {Vaughan}}]{Noy84}
{Noyes}, R.~W., {Hartmann}, L.~W., {Baliunas}, S.~L., {Duncan}, D.~K., \&
  {Vaughan}, A.~H. 1984, \apj, 279, 763

\bibitem[{{N{\'u}{\~n}ez} {et~al.}(2015){N{\'u}{\~n}ez}, {Ag{\"u}eros},
  {Covey}, {Hartman}, {Kraus}, {Bowsher}, {Douglas}, {L{\'o}pez-Morales},
  {Pooley}, {Posselt}, {Saar}, \& {West}}]{Nunez2015}
{N{\'u}{\~n}ez}, A., {Ag{\"u}eros}, M.~A., {Covey}, K.~R., {et~al.} 2015, \apj,
  809, 161

\bibitem[{{Nutzman} \& {Charbonneau}(2008)}]{Nut08}
{Nutzman}, P., \& {Charbonneau}, D. 2008, \pasp, 120, 317

\bibitem[{{Owen} \& {Mohanty}(2016)}]{Owe16}
{Owen}, J.~E., \& {Mohanty}, S. 2016, \mnras, 459, 4088

\bibitem[{{Pallavicini} {et~al.}(1981){Pallavicini}, {Golub}, {Rosner},
  {Vaiana}, {Ayres}, \& {Linsky}}]{Pal81}
{Pallavicini}, R., {Golub}, L., {Rosner}, R., {et~al.} 1981, \apj, 248, 279

\bibitem[{{Parker}(1955)}]{Par55}
{Parker}, E.~N. 1955, \apj, 122, 293

\bibitem[{{Parker}(1975)}]{Par75}
---. 1975, \apj, 198, 205

\bibitem[{{Perdelwitz} {et~al.}(2021){Perdelwitz}, {Mittag}, {Tal-Or},
  {Schmitt}, {Caballero}, {Jeffers}, {Reiners}, {Schweitzer}, {Trifonov},
  {Ribas}, {Quirrenbach}, {Amado}, {Seifert}, {Cifuentes},
  {Cort{\'e}s-Contreras}, {Montes}, {Revilla}, \&
  {Skrzypinski}}]{Perdelwitz2021}
{Perdelwitz}, V., {Mittag}, M., {Tal-Or}, L., {et~al.} 2021, \aap, 652, A116

\bibitem[{{Pizzolato} {et~al.}(2000){Pizzolato}, {Maggio}, {Micela},
  {Sciortino}, {Ventura}, \& {D'Antona}}]{Pizzolato2000}
{Pizzolato}, N., {Maggio}, A., {Micela}, G., {et~al.} 2000, in Astronomical
  Society of the Pacific Conference Series, Vol. 198, Stellar Clusters and
  Associations: Convection, Rotation, and Dynamos, ed. R.~{Pallavicini},
  G.~{Micela}, \& S.~{Sciortino}, 71

\bibitem[{{Rauscher} \& {Marcy}(2006)}]{Rauscher2006}
{Rauscher}, E., \& {Marcy}, G.~W. 2006, \pasp, 118, 617

\bibitem[{{Reiners} {et~al.}(2014){Reiners}, {Sch{\"u}ssler}, \&
  {Passegger}}]{Reiners2014}
{Reiners}, A., {Sch{\"u}ssler}, M., \& {Passegger}, V.~M. 2014, \apj, 794, 144

\bibitem[{Robertson {et~al.}(2014)Robertson, Mahadevan, Endl, \& Roy}]{Rob14}
Robertson, P., Mahadevan, S., Endl, M., \& Roy, A. 2014, Science, 345, 440

\bibitem[{{Rutten}(1984)}]{Rutten1984}
{Rutten}, R.~G.~M. 1984, \aap, 130, 353

\bibitem[{Salvatier {et~al.}(2016)Salvatier, Wiecki, \&
  Fonnesbeck}]{Salvatier2016}
Salvatier, J., Wiecki, T.~V., \& Fonnesbeck, C. 2016, {PeerJ} Computer Science,
  2, e55.
\newblock \url{https://doi.org/10.7717/peerj-cs.55}

\bibitem[{{Schmidt} {et~al.}(2014){Schmidt}, {Prieto}, {Stanek}, {Shappee},
  {Morrell}, {Bardalez Gagliuffi}, {Kochanek}, {Jencson}, {Holoien}, {Basu},
  {Beacom}, {Szczygie{\l}}, {Pojmanski}, {Brimacombe}, {Dubberley}, {Elphick},
  {Foale}, {Hawkins}, {Mullins}, {Rosing}, {Ross}, \& {Walker}}]{Sch14}
{Schmidt}, S.~J., {Prieto}, J.~L., {Stanek}, K.~Z., {et~al.} 2014, \apjl, 781,
  L24

\bibitem[{{Schrijver} \& {Rutten}(1987)}]{Schrijver1987}
{Schrijver}, C.~J., \& {Rutten}, R.~G.~M. 1987, \aap, 177, 143

\bibitem[{{Schr{\"o}der} {et~al.}(2009){Schr{\"o}der}, {Reiners}, \&
  {Schmitt}}]{Schroder2009}
{Schr{\"o}der}, C., {Reiners}, A., \& {Schmitt}, J.~H.~M.~M. 2009, \aap, 493,
  1099

\bibitem[{{Shields} {et~al.}(2016){Shields}, {Ballard}, \& {Johnson}}]{Shi16}
{Shields}, A.~L., {Ballard}, S., \& {Johnson}, J.~A. 2016, \physrep, 663, 1

\bibitem[{{Skumanich}(1972)}]{Skumanich1972}
{Skumanich}, A. 1972, \apj, 171, 565

\bibitem[{{Su{\'a}rez Mascare{\~n}o} {et~al.}(2016){Su{\'a}rez Mascare{\~n}o},
  {Rebolo}, \& {Gonz{\'a}lez Hern{\'a}ndez}}]{SuarezMascareno2016}
{Su{\'a}rez Mascare{\~n}o}, A., {Rebolo}, R., \& {Gonz{\'a}lez Hern{\'a}ndez},
  J.~I. 2016, \aap, 595, A12

\bibitem[{{Su{\'a}rez Mascare{\~n}o} {et~al.}(2015){Su{\'a}rez Mascare{\~n}o},
  {Rebolo}, {Gonz{\'a}lez Hern{\'a}ndez}, \& {Esposito}}]{Sua15}
{Su{\'a}rez Mascare{\~n}o}, A., {Rebolo}, R., {Gonz{\'a}lez Hern{\'a}ndez},
  J.~I., \& {Esposito}, M. 2015, \mnras, 452, 2745

\bibitem[{{Thompson} {et~al.}(1996){Thompson}, {Toomre}, {Anderson}, {Antia},
  {Berthomieu}, {Burtonclay}, {Chitre}, {Christensen-Dalsgaard}, {Corbard}, {De
  Rosa}, {Genovese}, {Gough}, {Haber}, {Harvey}, {Hill}, {Howe}, {Korzennik},
  {Kosovichev}, {Leibacher}, {Pijpers}, {Provost}, {Rhodes}, {Schou}, {Sekii},
  {Stark}, \& {Wilson}}]{Tho96}
{Thompson}, M.~J., {Toomre}, J., {Anderson}, E.~R., {et~al.} 1996, Science,
  272, 1300

\bibitem[{{Tomczyk} {et~al.}(1996){Tomczyk}, {Schou}, \& {Thompson}}]{Tom96}
{Tomczyk}, S., {Schou}, J., \& {Thompson}, M.~J. 1996, in American Astronomical
  Society Meeting Abstracts, Vol. 188, American Astronomical Society Meeting
  Abstracts \#188, 69.03

\bibitem[{{Torres} {et~al.}(2006){Torres}, {Quast}, {da Silva}, {de La Reza},
  {Melo}, \& {Sterzik}}]{Tor06}
{Torres}, C.~A.~O., {Quast}, G.~R., {da Silva}, L., {et~al.} 2006, \aap, 460,
  695

\bibitem[{Vanderburg {et~al.}(2016)Vanderburg, Plavchan, Johnson, Ciardi,
  Swift, \& Kane}]{Van16}
Vanderburg, A., Plavchan, P., Johnson, J.~A., {et~al.} 2016, Monthly Notices of
  the Royal Astronomical Society, 459, 3565

\bibitem[{{Vaughan} {et~al.}(1981){Vaughan}, {Baliunas}, {Middelkoop},
  {Hartmann}, {Mihalas}, {Noyes}, \& {Preston}}]{Vau81}
{Vaughan}, A.~H., {Baliunas}, S.~L., {Middelkoop}, F., {et~al.} 1981, \apj,
  250, 276

\bibitem[{Vaughan \& Preston(1980)}]{vaughan1980}
Vaughan, A.~H., \& Preston, G.~W. 1980, Publications of the Astronomical
  Society of the Pacific, 92, 385

\bibitem[{{Vilhu}(1984)}]{Vilhu1984}
{Vilhu}, O. 1984, \aap, 133, 117

\bibitem[{{West} {et~al.}(2015){West}, {Weisenburger}, {Irwin},
  {Berta-Thompson}, {Charbonneau}, {Dittmann}, \& {Pineda}}]{West2015}
{West}, A.~A., {Weisenburger}, K.~L., {Irwin}, J., {et~al.} 2015, \apj, 812, 3

\bibitem[{{Wilson}(1963)}]{Wil63}
{Wilson}, O.~C. 1963, \apj, 138, 832

\bibitem[{{Wilson}(1968)}]{Wil68}
---. 1968, \apj, 153, 221

\bibitem[{Wright {et~al.}(2011)Wright, Drake, Mamajek, \& Henry}]{Wri11}
Wright, N.~J., Drake, J.~J., Mamajek, E.~E., \& Henry, G.~W. 2011, The
  Astrophysical Journal, 743, 48.
\newblock \url{https://doi.org/10.1088%2F0004-637x%2F743%2F1%2F48}

\bibitem[{{Wright} {et~al.}(2018){Wright}, {Newton}, {Williams}, {Drake}, \&
  {Yadav}}]{Wri18}
{Wright}, N.~J., {Newton}, E.~R., {Williams}, P. K.~G., {Drake}, J.~J., \&
  {Yadav}, R.~K. 2018, \mnras, 479, 2351

\end{thebibliography}

\section{Observations \& Data Reduction}\label{sec:Observations}
We initially selected a sample of 55 mid-late M-dwarfs from targets of
the MEarth survey \citep{Ber12} to observe. Targets were selected based on high
proper motions and availability of a previously measured photometric rotation
period, or an expectation of a measurement based on data available from
MEarth-South at the time. These rotational periods were derived photometrically
\citep[e.g.][]{Newton2016,Man16,Med20}. For star 2MASS J06022261-2019447, which
was categorized as an ``uncertain detection'' from MEarth photometry by
\citet{Newton2018}, including new data from MEarth DR10 we find a period of 95
days. This value was determined following similar methodology to \citet{Irw11}
and \citet{Newton2016,Newton2018}, and is close to the reported candidate
period of 116 days.  References for all periods are provided in the machine
readable version of Table \ref{tab:finalData}.   

High resolution spectra were collected from March to October 2017 using the
Magellan Inamori Kyocera Echelle (MIKE) spectrograh on the 6.5 meter Magellan 2
telescope at the Las Campanas Observatory in Chile. MIKE is a high resolution
double echelle spectrograph with blue and red arms. Respectively, these cover
wavelengths from 3350 - 5000 \AA\ and 4900-9500 \AA\ \citep{Ber03}. We
collected data using a 0.75x5.00" slit resulting in a resolving power of 32700.
Each science target was observed an average of four times with mean integration
times per observation ranging from 53.3 to 1500 seconds. Ca II H\&K
lines were observed over a wide range of signa-to-noise ratios, from $\sim 5$ up
to $\sim 240$ with mean and median values of 68 and 61 respectively.

We use the \texttt{CarPy} pipeline \citep{Kel00, Kel03} to reduce our blue arm
spectra. \texttt{CarPy}'s data products are wavelength calibrated, blaze
corrected, and background subtracted spectra comprising 36 orders. We shift all
resultant target spectra into the rest frame by cross correlating against a
velocity template spectrum. For the velocity template we use an observation of
Proxima Centari in our sample. This spectrum's velocity is both barycentrically
corrected, using astropy's \texttt{SkyCoord} module \citep{Ast18}, and
corrected for Proxima Centari's measured radial velocity, -22.4 km s$^{-1}$
\citep{Tor06}. Each echelle order of every other target observation is cross
correlated against the corresponding order in the template spectra using
\texttt{specutils} \texttt{template\_correlate} function \citep{Nic21}.
Velocity offsets for each order are inferred from a Gaussian fit to the
correlation vs. velocity lag function. For each target, we apply a three sigma
clip to list of echelle order velocities, visually verifying this clip removed
low S/N orders. We take the mean of the sigma-clipped velocities Finally, each
wavelength bin is shifted according to its measured velocity.

Ultimately, two targets (2MASS J16570570-0420559 and 2MASS J04102815-5336078)
had S/N ratios around the Ca II H\&K lines which were too low to be of use,
reducing the number of R'$_{HK}$ measurement we can make from 55 to 53. 
